Author’s response to reviews

Title: Computerized clinical decision support system for diabetes in primary care does not improve quality of care: a cluster-randomized controlled trial

Authors:

Annemie Heselmans (Annemie.heselmans@gmail.com;annemie.heselmans@kuleuven.be)
Nicolas Delvaux (nicolas.delvaux@kuleuven.be)
Annouschka Laenen (annouschka.laenen@kuleuven.be)
Stijn Van de Velde (Stijn.vandevelde@fhi.no)
Dirk Ramaekers (Dirk.Ramaekers@kuleuven.be)
Ilkka Kunnamo (Ilkka.Kunnamo@duodecim.fi)
Bert Aertgeerts (Bert.Aertgeerts@kuleuven.be)

Version: 2 Date: 22 Nov 2019

Author’s response to reviews:

Dear editor,

We wish to submit a revised manuscript entitled ‘Computerized clinical decision support system for diabetes in primary care does not improve quality of care: a cluster randomized controlled trial’. A copy editor went now through the manuscript and the length of the discussion section was cut by about 20% (from 2743 words to 2168). We attached a copy with track changes and a clean copy which is easier to read.

We hope that the changes we have made have further improved the manuscript.

Yours sincerely,

Annemie Heselmans